Author's response to reviews

Title: A psychometric appraisal of the DREEM

Authors:

Sean Hammond (s.hammond@ucc.ie)
Margaret O'Rourke (m.orourke@ucc.ie)
Martina Kelly (m.kelly@ucc.ie)
Deirdre Bennett (d.bennett@ucc.ie)
Siun O'Flynn (siun.oflynn@ucc.ie)

Version: 4 Date: 6 January 2012

Author's response to reviews:

We have made the required changes to the title page and have inserted the authors details as requested.

The Abstract has been added to the text as requested.

The referencing has been checked and adjusted to conform to the format required.

Competing interest section has been added.

Authors contribution section has been added.